Author's response to reviews

Title: Cost-effectiveness of a participatory return-to-work intervention for temporary agency workers and unemployed workers sick-listed due to musculoskeletal disorders: design of a randomised controlled trial.

Authors:

Sylvia J Vermeulen (s.vermeulen@vumc.nl)
Johannes R Anema (j.anema@vumc.nl)
Antonius JM Schellart (ton.schellart@vumc.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)
Allard J van der Beek (a.vanderbeek@vumc.nl)

Version: 3 Date: 17 February 2010

Author's response to reviews:

Dear editor,

My apologies for the delayed response.

After reading the manuscript one final time I discovered a few typographical and (minor) content errors. These have been corrected. In addition, table 2 has been corrected (content of column with measurements at 9 months was missing) and figure 1 contained a typographical error. This has been corrected.

Again, my apologies for any inconvenience.

With kind regards,

Sylvia J. Vermeulen, MD, MSc
Research Centre for Insurance Medicine AMC-UWV-VUmc
EMGO Institute for Health and Care Research – VU University Medical Centre
P.O. Box 7057
1007 MB Amsterdam
The Netherlands